Author's response to reviews

Title: Exclusive breastfeeding of Swedish children and its possible influence on development of obesity: a prospective cohort study.

Authors:

Karina Huus (huca@hhj.hj.se)
Jonas F Ludvigsson (jonasludvigsson@yahoo.com)
Karin Enskar (Karin.Enskar@hhi.hj.se)
Johnny Ludvigsson (Johnny.Ludvigsson@lio.se)

Version: 8 Date: 2 October 2008

Author's response to reviews: see over
To be re-submitted to BMC Pediatrics

MS: 1200503532168972

Jönköping, Sweden, 2008-10-02

About the formatting changes requested for our manuscript "Exclusive breastfeeding of Swedish children and possible influence on development of obesity: a prospective cohort study". According to your suggestions we have change the title and the commas in numbers. We have also removed the keywords, word- counts and short title. About the reference list I send a mail to you and discussed it and now it should be right because I used the journal´s template in Endnote. I hope that this will be good please let me know if there is anything else you want me to do.

Sincerely,

Karina Huus, RN, PhD student
Jonas F. Ludvigsson, MD, PhD

Karin. Enskär,RN, PhD  
Johnny Ludvigsson, MD, PhD

Corresponding address:
Karina Huus
Department of Nursing Science
School of Health Science, Jönköping University

P.O.Box 1026, S-551 11 Jönköping

Sweden

Tel: +46 36 10 12 34 Fax:+46 36 10 12 50

E-mail: karina.huus@hhj.hj.se
Dear reviewer: Wendy H Oddy

Thank you very much for having revised our manuscript once again. We have made the grammatical changes as you suggested. But I have not change the referencing format because according to how I understand it from the instructions it should be a colon following authors names in journal references, please let me know if it is something I haven't understand.

/Karina Huus